Author's response to reviews

Title: Prevalence of frailty and its associated factors in older hospitalised patients in Vietnam

Authors:
Huyen Vu (vuthanhhuyen11@hmu.edu.vn)
Thanh Nguyen (xuanthanhl901vlk@gmail.com)
Tu Nguyen (nntu81@gmail.com)
Anh Nguyen (trunganhvlk@gmail.com)
Robert Cumming (robert.cumming@sydney.edu.au)
Sarah Hilmer (sarah.hilmer@sydney.edu.au)
Thang Pham (pham_thang-vlk@yahoo.com)

Version: 3 Date: 22 Aug 2017

Author’s response to reviews:

Dear Editors,

We would like to thank you for the time spent assessing our manuscript and for the opportunity to respond.

We have specifically responded to the issues raised as detailed below, and would be happy to address any further issues if required.

1. Please note that all manuscripts must contain all the following sections under the heading 'Declarations'. The Declarations should follow the Conclusions section, and be before the References.

Abbreviations

Ethics approval and consent to participate

Consent for publication

Availability of data and material
Competing interests

Funding

Authors' contributions

Acknowledgements

Please see here for details on the information to be included in these sections:

https://bmcgeriatr.biomedcentral.com/submission-guidelines/preparing-your-manuscript/research-article

If the information required is already provided in the main manuscript, please also copy the relevant statements to the Declarations.

If any of the sections are not relevant to your manuscript, please include the heading and write 'Not applicable' for that section.

Response: We have added these sections in the manuscript as advised.

2. Please include a Keywords section after the Abstract, with three to ten keywords related to this study.

Response: We have added 4 keywords after the Abstract.

3. Please change your Introduction heading to Background.

Response: We have changed "Introduction" heading to "Background".

4. Please remove the Aim heading from the Abstract, this section is not included in Abstracts in BMC Geriatrics.

Response: We have removed the "Aim" heading from the Abstract.

5. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

Response: We have prepared a clean manuscript as advised.